Author's response to reviews

Title: Ectopic pancreas tissue appearing in a mediastinal cyst

Authors:

Sandor Szabados (sandor.szabados@aok.pte.hu)
Laszlo Lenard (laszlo_11_lenard@yahoo.com)
Tamas Tornoczky (tamas.tornoczky@aok.pte.hu)
Edit Varady (edit.varady@aok.pte.hu)
Zsofia Verzár (zsofia.verzar@kk.pte.hu)

Version: 4 Date: 14 February 2012

Author's response to reviews: see over
Dear Editor,

We would like to submit a very interesting and also extremely rare case with picturesque documentation in your journal, as a case report.

The title of our paper is: Ectopic pancreas tissue appearing in a mediastinal cyst.

Heterotopia of pancreatic tissue is a common developmental anomaly. Ectopic pancreatic tissue is mostly found in the gastrointestinal tract, localization in the mediastinum is extremely rare. We have found only six such cases in the English literature.

The authors profess the following statements: we have not submitted this paper in any other journal, all authors have read and approved the manuscript, the authors will transfer copyright to the Publisher, and there is no conflict of interest.

Name of authors: Sándor Szabados
László Lénárd, Tamás Tornóczky, Edit Várady, Zsófia Verzár,*

Institution: University of Pécs, Heart Clinic, Pécs, Hungary

* University of Pécs, Anesthesiology and Intensive Care, Pécs Hungary.

Corresponding author:
Sandor Szabados M.D., Ph.D.

Tel: 00 36 30 2770658

Fax: 00 36 72 536 000/5658

Address: Heart Clinic, Ifjúság. u. 13. 7624, Pécs, Hungary

E-mail: sandor.szabados@aok.pte.hu.
Sincerely yours,

Laszlo Lenard